Author's response to reviews

Title: Bacteriophage-based tests for the detection of Mycobacterium tuberculosis in clinical specimens: a systematic review and meta-analysis

Authors:

- Shriprakash Kalantri (kalantri@berkeley.edu)
- Madhukar Pai (madhupai@berkeley.edu)
- Lisa Pascopella (Pascope@dhs.ca.gov)
- Lee Riley (lwriley@berkeley.edu)
- Arthur Reingold (reingold@berkeley.edu)

Version: 3 Date: 15 July 2005
Dear Editor:

Many thanks for your decision e-mail (13 July 2005), regarding our manuscript MS: 2094516714620543. We are happy that you found that we have addressed reviewer’s criticisms satisfactorily. We are delighted that you have accepted our revised manuscript.

As requested, we have deleted keywords section, removed page breaks, highlighting and tracking. We have also ensured that the image files do not include titles and legends. We have added a page after the reference list that includes description of titles and legends.

I am uploading the following documents with this cover note:

1. One electronic copy of the revised manuscript (complete, including tables, figures, references, acknowledgements and author contributions).

2. Title, abstract and author details: We have included country in the affiliation details.

3. Figures

4. Permission note to use figure 1 from *Biomedica*

Thanking you,

Sincerely,

Dr. Shripakash Kalantri, MD, MPH